Author's response to reviews

Title: Treatment approaches of recurrent anastomotic strictures following artificial urinary sphincter implantation. A case presentation.

Authors:

Ioannis Adamakis (yanton@hotmail.com)
Stavros I. Tyritzis (statyr@freemail.gr)
Ioanna Vasileiou (tzovasileiou@hotmail.com)
Ioannis Katafigiotis (katafigiotis@yahoo.com)
Ioannis Leotsakos (leotsakos@gmail.com)
Sotiria Fergadaki (sotiria22@yahoo.com)
Konstantinos G. Stravodimos (kgstravod@yahoo.com)
Constantinos A. Constantinides (ckonstan@med.uoa.gr)

Version: 2 Date: 2 November 2011

Author's response to reviews:

we proceeded to the changes in the format according to the guidelines for authors, as requested.